Author's response to reviews

Title: Degranulating mast cells in fibrotic regions of human tumors and evidence that mast cell heparin interferes with the growth of tumor cells through a mechanism involving fibroblasts.

Authors:  
  Michael K Samoszuk (msamoszu@uci.edu)  
  Emi Kanakubo (emik@uci.edu)  
  John K Chan (johnchan@stanford.edu)

Version: 3  Date: 16 September 2005

Author's response to reviews:

To the Editor:
We have made all of the formatting changes that were requested in your e-mail message of September 15, 2005. Specifically, we ensured that the title and abstract are identical in the manuscript and submission system. We also modified the reference list to include the names of all authors and deleted the use of "et al." In addition, we verified the correct order and list of all authors in the submission system and manuscript.

Thank you for accepting our paper for publication.

Sincerely,

Michael Samoszuk